Author's response to reviews

Title: Correlation of three immunohistochemically detected markers of neuroendocrine differentiation with clinical predictors of disease progression in prostate cancer

Authors:

M Hammad Ather (hammad.ather@aku.edu)
Abbas Farhat (farhat.abbas@aku.edu)
Faruqui Nuzhat (nuzhat.faruqui@aku.edu)
Israr Mohammad (mohammad.israr@aku.edu)
Pervez Shahid (shahid.pervez@aku.edu)

Version: 5 Date: 19 December 2008

Author's response to reviews: see over
Formatting changes requested
----------------------------------

Major revisions (we require the author to make these changes)

Affiliations - Each affiliation should be written in full, in the following format: department, institute, city, and country. This may mean repeating institutional addresses if the authors are from different departments within an institution
Authors’ reply: done

Table vertical elements - Please note that we are unable to display vertical rules within the tables nor are we able to write text vertically within the cells - please re-layout your table without these vertical elements. We are unable to correctly display merged cells where the merged cell crosses rows.
Authors’ reply: done, tables re done and uploaded as separate file

Minor revisions

Email address - please include email addresses for all the authors on the title page. Please use the following format-
GSF: g.s.feder@qmul.ac.uk
HS: hsohal@yahoo.co.uk
SE: s.eldridge@qmul.ac.uk
Authors’ reply: done

Please change 'Introduction' to 'Background'.

Please change 'Materials and methods' to 'Methods'.

Please format Authors' contributions as a paragraph rather than a list.

Add the heading 'Tables' to the table section.
Authors’ reply: done

Thank you.

Dr M H Ather